Novartis initiates study evaluating impact of higher dosing of Cosentyx® in patients with ankylosing spondylitis

Dec 07, 2017

- - ASLeap trial to evaluate impact of increasing the dose of Cosentyx (secukinumab) to 300 mg in patients who do not achieve symptom remission after 16 weeks of treatment with the approved dose of 150 mg
- - Additional exploratory analysis to investigate sleep disturbance, fatigue and daytime activity, some of the most common concerns for patients with ankylosing spondylitis(1, 2)
- - Enrollment expected to begin January 2018(1)

EAST HANOVER, N.J., Dec. 7, 2017 /PRNewswire/ -- Novartis announced today the initiation of the ASLeap trial in patients with ankylosing spondylitis (AS), evaluating the effect of changing to a higher dose (300 mg) of Cosentyx[®] (secukinumab) in patients who do not achieve symptom remission after treatment with Cosentyx 150 mg for 16 weeks. The primary endpoint of ASLeap is to determine the difference between Cosentyx 300 mg and Cosentyx 150 mg at Week 52 based on the proportion of subjects achieving inactive disease status based on the Ankylosing Spondylitis Disease Activity Score (ASDAS).¹

"An important goal of ankylosing spondylitis treatment is to provide as much symptom relief as possible for patients living with this debilitating disease," said Marcia Kayath, Head US Clinical Development and Medical Affairs, Novartis Pharmaceuticals Corporation. "While Cosentyx 150 mg has shown clinically significant results in treating a majority of patients with AS, we want to investigate whether some patients may benefit from a higher dose. We hope that the results of the ASLeap study will help provide physicians with important information about how best to manage these patients."

An exploratory analysis will also assess sleep disturbance and daytime activity in patients with AS, as measured by the use of a wearable motion biosensor, an Actiwatch[®]* device, which is a medical device resembling a wristwatch used to collect detailed information on sleep and physical activity. Patients with AS report chronic and extensive sleep disturbance due to pain and stiffness during the night.² In AS patients, poor quality sleep is strongly correlated with increased pain, fatigue, lower quality of life, higher depressed mood, higher disease activity and reduced physical function.³

The study is expected to begin enrollment in January 2018 and aims to enroll approximately 270 patients at 70 centers in the United States. More information can be found at www.clinicaltrials.gov.

About the ASLeap Study

The ASLeap study design includes an initial 16 week open-label period (Treatment Period 1), followed by a randomized, double-blind, parallel-group period (Treatment Period 2). During the initial 16 weeks of the trial, all patients will receive open-label 150 mg Cosentyx. At Week 16, based on whether patients have met the definition of ASDAS inactive disease, patients will be placed into one of the following three groups for the remaining 36 weeks of the trial: Responders, patients who achieve ASDAS inactive disease (total score <1.3) at both Week 12 and Week 16, will enter Treatment Period 2 and continue receiving blinded Cosentyx 150 mg through Week 48. Inadequate Responders, patients who are not in inactive disease status, defined as an ASDAS total score of ≥1.3, at both Week 12 and Week 16, will enter Treatment Period 2 and be randomized

(1:1, double-blinded) to Cosentyx 300 mg or Cosentyx 150 mg through Week 48. Non-responders, patients who exhibit no change or an increase (worsening) from baseline in total ASDAS score at either Week 12 or Week 16, will complete the study at Week 16.

The study population will consist of male and female patients, ≥18 years of age with a clinical diagnosis of moderate to severe AS fulfilling the Modified New York criteria for AS despite previous or current NSAIDs/non-biologic DMARDs and/or anti-TNFα therapy. Patients must have active AS, as measured by the following: total BASDAI ≥4 on a scale of 0-10 at baseline, spinal pain as measured by BASDAI question #2 ≥4 on a scale of 0-10 at baseline, and total back pain as measured by visual analog scale (VAS) ≥40mm (0-100 mm) at baseline. ¹

About ankylosing spondylitis (AS)

Ankylosing spondylitis (AS) is a painful and often progressively debilitating disease, caused by spine inflammation that can result in irreversible damage. Up to 70% of patients who go on to develop severe AS will form spinal fusions (where the bones grow together) over 10 to 15 years, which significantly reduces mobility. People in their teens and twenties, particularly males, are affected most often. Family members of those with AS are at higher risk. Approximately 20-40% of patients do not respond well to standard of care biologic medicines, and there have been few therapeutic options for those people.

About Cosentyx (secukinumab) and interleukin-17A (IL-17A)
Cosentyx is a fully human monoclonal antibody (mAB) that selectively binds to the interleukin-17A (IL-17A) cytokine and inhibits its interaction with the IL-17 receptor.^{8,9}

Cosentyx is the only IL-17a antagonist approved for the treatment of adults with active ankylosing spondylitis. Cosentyx is approved in more than 70 countries, which includes the European Union countries and the US. Cosentyx is also approved for the treatment of PsA and pustular psoriasis in Japan.¹⁰

In addition, Cosentyx is approved in more than 75 countries for the treatment of moderate to severe plaque psoriasis, which includes the European Union countries, Japan, Switzerland, Australia, the US and Canada. In Europe, Cosentyx is approved for the first-line systemic treatment of moderate to severe plaque psoriasis in adult patients.¹¹ In the US, Cosentyx is approved as a treatment for moderate to severe plaque psoriasis in adult patients who are candidates for systemic therapy or phototherapy (light therapy).⁹

To date, more than 100,000 patients worldwide have been prescribed Cosentyx in the post-marketing setting across all indications. ¹² In addition, 2017 marks 10 years since the first patient visit in a clinical trial with Cosentyx. ¹²

INDICATIONS

Cosentyx is a human interleukin-17A antagonist indicated for the treatment of:

- moderate to severe plaque psoriasis in adult patients who are candidates for systemic therapy or phototherapy
- adults with active psoriatic arthritis (PsA)
- adults with active ankylosing spondylitis (AS)

The approved dosage for the treatment of AS is 150 mg.

IMPORTANT SAFETY INFORMATION

Do not use Cosentyx if you have had a severe allergic reaction to secukinumab or any of the other ingredients

in Cosentyx. See the Medication Guide for a complete list of ingredients.

Cosentyx is a medicine that affects your immune system. Cosentyx may increase your risk of having serious side effects such as:

Infections

Cosentyx may lower the ability of your immune system to fight infections and may increase your risk of infections.

- Your doctor should check you for tuberculosis (TB) before starting treatment with Cosentyx.
- If your doctor feels that you are at risk for TB, you may be treated with medicine for TB before you begin treatment with Cosentyx and during treatment with Cosentyx.
- Your doctor should watch you closely for signs and symptoms of TB during treatment with Cosentyx. Do not take Cosentyx if you have an active TB infection.

Before starting Cosentyx, tell your doctor if you:

- are being treated for an infection
- have an infection that does not go away or that keeps coming back
- have TB or have been in close contact with someone with TB
- think you have an infection or have symptoms of an infection such as:
- fevers, sweats, or chills warm, red, or painful skin or
- muscle aches sores on your body
- cough diarrhea or stomach pain
- shortness of breath burning when you urinate
- blood in your phlegm or urinate more often than
- weight loss normal

After starting Cosentyx, call your doctor right away if you have any signs of infection listed above. Do not use Cosentyx if you have any signs of infection unless you are instructed to by your doctor.

Inflammatory Bowel Disease

New cases of inflammatory bowel disease or "flare-ups" can happen with Cosentyx, and can sometimes be serious. If you have inflammatory bowel disease (ulcerative colitis or Crohn's disease), tell your doctor if you have worsening disease symptoms during treatment with Cosentyx or develop new symptoms of stomach pain or diarrhea.

Serious Allergic Reactions

Serious allergic reactions can occur. Get emergency medical help right away if you get any of the following

symptoms: feeling faint; swelling of your face, eyelids, lips, mouth, tongue, or throat; trouble breathing or throat tightness; chest tightness; or skin rash. If you have a severe allergic reaction, do not give another injection of Cosentyx.

Before starting Cosentyx, tell your doctor if you:

- have any of the conditions or symptoms listed above for infections
- have inflammatory bowel disease (Crohn's disease or ulcerative colitis)
- are allergic to latex. The needle caps contain latex.
- have recently received or are scheduled to receive an immunization (vaccine). People who take Cosentyx should not receive live vaccines.
- have any other medical conditions
- are pregnant or plan to become pregnant. It is not known if Cosentyx can harm your unborn baby. You and your doctor should decide if you will use Cosentyx.
- are breastfeeding or plan to breastfeed. It is not known if Cosentyx passes into your breast milk.

Tell your doctor about all the medicines you take, including prescription and over-the-counter medicines, vitamins, and herbal supplements. Know the medicines you take. Keep a list of your medicines to show your doctor and pharmacist when you get a new medicine.

How should I use Cosentyx?

See the detailed Instructions for Use that comes with your Cosentyx for information on how to prepare and inject a dose of Cosentyx, and how to properly throw away (dispose of) used Cosentyx Sensoready® pens and prefilled syringes.

- Use Cosentyx exactly as prescribed by your doctor.
- If your doctor decides that you or a caregiver may give your injections of Cosentyx at home, you should receive training on the right way to prepare and inject Cosentyx. Do not try to inject Cosentyx yourself, until you or your caregiver has been shown how to inject Cosentyx by your doctor or nurse.

The most common side effects of Cosentyx include: cold symptoms, diarrhea, and upper respiratory infections. These are not all of the possible side effects of Cosentyx. Call your doctor for medical advice about side effects.

You are encouraged to report negative side effects of prescription drugs to the FDA. Visit www.fda.gov/medwatch, or call 1-800-FDA-1088.

Please see accompanying full <u>Prescribing Information</u>, including <u>Medication Guide</u>.

Disclaimer

This press release contains forward-looking statements within the meaning of the United States Private Securities Litigation Reform Act of 1995. Forward-looking statements can generally be identified by words such as "potential," "can," "will," "plan," "expect," "anticipate," "look forward," "believe," "committed," "investigational," "pipeline," "launch," or similar terms, or by express or implied discussions regarding potential marketing approvals, new indications or labeling for the investigational or approved products described in this press release, or regarding potential future revenues from such products. You should not place undue reliance on these statements. Such forward-looking statements are based on our current beliefs and expectations regarding future events, and are subject to significant known and unknown risks and uncertainties. Should one or more of these risks or uncertainties materialize, or should underlying assumptions prove incorrect, actual results may vary materially from those set forth in the typ ward-looking statements. There can be no guarantee

that the investigational or approved products described in this press release will be submitted or approved for sale or for any additional indications or labeling in any market, or at any particular time. Nor can there be any guarantee that such products will be commercially successful in the future. In particular, our expectations regarding such products could be affected by, among other things, the uncertainties inherent in research and development, including clinical trial results and additional analysis of existing clinical data; regulatory actions or delays or government regulation generally; our ability to obtain or maintain proprietary intellectual property protection; the particular prescribing preferences of physicians and patients; global trends toward health care cost containment, including government, payor and general public pricing and reimbursement pressures; general economic and industry conditions, including the effects of the persistently weak economic and financial environment in many countries; safety, quality or manufacturing issues, and other risks and factors referred to in Novartis AG's current Form 20-F on file with the US Securities and Exchange Commission.

Novartis is providing the information in this press release as of this date and does not undertake any obligation to update any forward-looking statements contained in this press release as a result of new information, future events or otherwise.

About Novartis

Located in East Hanover, NJ Novartis Pharmaceuticals Corporation is an affiliate of Novartis which provides innovative healthcare solutions that address the evolving needs of patients and societies. Headquartered in Basel, Switzerland, Novartis offers a diversified portfolio to best meet these needs: innovative medicines, cost-saving generic and biosimilar pharmaceuticals and eye care. Novartis has leading positions globally in each of these areas. In 2016, the Group achieved net sales of USD 48.5 billion, while R&D throughout the Group amounted to approximately USD 9.0 billion. Novartis Group companies employ approximately 121,000 full-time-equivalent associates. Novartis products are sold in approximately 155 countries around the world. For more information, please visit http://www.novartis.com.

Novartis is on Twitter. Sign up to follow @Novartis at http://twitter.com/novartis
For Novartis multimedia content, please visit www.novartis.com/news/media-library
For questions about the site or required registration, please contact media.relations@novartis.com

 * Actiwatch $^{\! (\! R \!)}$ is a registered trademark of Mini-Mitter Co. Inc.

References

- 1. Novartis Data on file. Clinical Trial Protocol CAIN457FUS06. November 2017.
- 2. Leverment S, Clarke E, Wadeley A, Sengupta R. Prevalence and factors associated with disturbed sleep in patients with ankylosing spondylitis and non-radiographic axial spondyloarthritis: a systematic review. Rheumatology International. 2016;37(2):257-271.
- 3. Batmaz I, Sarıyıldız MA, Dilek B, Bez Y, Karakoç M, Çevik R. Sleep quality and associated factors in ankylosing spondylitis: relationship with disease parameters, psychological status and quality of life. Rheumatology International. 2012;33(4):1039-1045.
- 4. Schett et al. Structural damage in rheumatoid arthritis, psoriatic arthritis, and ankylosing spondylitis: traditional views, novel insights gained from TNF blockade, and concepts for the future. Arthritis Res Ther. 2011;13:S4.
- 5. Barkham N, Kong K O, Tennant A, et al. The unmet need for anti-tumour necrosis factor (anti-TNF) therapy in ankylosing spondylitis. Rheumatology. 2005;44:1277–1281.
- 6. Dean LE, Jones GT, MacDonald AG, et al. Global prevalence of ankylosing spondylitis. Rheumatology (Oxford). 2014;53(4):650-7.
- 7. Dougados M, Baeten D. Spondyloarthritis. Lancet. 2011; 377(9783):2127-37.
- 8. McInnes, IB, Mease, PJ, Kirkham, B, et al. Secukinumab, a human anti-interleukin-17A monoclonal antibody, in patients with psoriatic arthritis (FUTURE 2): a randomised, double-blind, placebo-controlled,

- phase 3 trial. The Lancet. 2015; 386, No. 9999; p1137-1146.
- 9. Cosentyx [prescribing information]. East Hanover, NJ: Novartis Pharmaceuticals Corp; 2017
- 10. Pharmaceuticals and Medical Devices Agency. Review Report. Available at: http://www.pmda.go.jp/files/000216877.pdf . Accessed August 10, 2017.
- Cosentyx Summary of Product Characteristics. Novartis Europharm Limited. Available at: http://www.ema.europa.eu/ema/index.jsp?
 http://www.ema.europa.eu/ema/index.jsp?
 http://www.ema.europa.eu/ema/index.jsp?
 http://www.ema.europa.eu/ema/index.jsp?
 http://www.ema.europa.eu/ema/index.jsp?
 http://www.ema.europa.eu/ema/index.jsp?
 http://www.ema.eu/ema/index.jsp?
 <a href="curl=pages/medicines/human/medicines/
- 12. Novartis Data on file. Number of Patients Prescribed Cosentyx. May 2017.

Novartis Media Relations

Central media line: +41 61 324 2200 E-mail: media.relations@novartis.com

Eric Althoff Jeannie Neufeld

Novartis Global Media Relations Novartis Pharmaceuticals Corporation

+41 61 324 7999 (direct) +1 862 778 2104 (direct)

+41 79 593 4202 (mobile) +1 201 650 2728 (mobile)

<u>eric.althoff@novartis.com</u> <u>jeannie.neufeld@novartis.com</u>

Novartis Investor Relations

Central investor relations line: +41 61 324 7944

E-mail: investor.relations@novartis.com

Central North America

Samir Shah +41 61 324 7944 Richard Pulik +1 212 830 2448

Pierre-Michel Bringer +41 61 324 1065 Cory Twining +1 212 830 2417

Thomas Hungerbuehler +41 61 324 8425

Isabella Zinck +41 61 324 7188

SOURCE Novartis Pharmaceuticals Corporation

Source URL: https://prod1.novartis.com/us-en/news/media-releases/novartis-initiates-study-evaluating-

List of links present in page

- 1. https://prod1.novartis.com/us-en/us-en/news/media-releases/novartis-initiates-study-evaluating-impact-higher-dosing-cosentyx-patients-ankylosing-spondylitis
- 2. http://www.clinicaltrials.gov/
- 3. http://www.fda.gov/medwatch
- 4. http://www.pharma.us.novartis.com/product/pi/pdf/cosentyx.pdf
- 5. http://www.pharma.us.novartis.com/product/pi/pdf/cosentyx_pmg.pdf
- 6. http://www.novartis.com/
- 7. http://twitter.com/novartis
- 8. http://www.novartis.com/news/media-library
- 9. mailto:media.relations@novartis.com
- 10. http://www.ncbi.nlm.nih.gov/pubmed/24324212
- 11. http://www.pmda.go.jp/files/000216877.pdf
- 12. http://www.ema.europa.eu/ema/index.jsp? curl=pages/medicines/human/medicines/003729/human_med_001832.jsp&mid=WC0b01ac058001d124
- 13. mailto:media.relations@novartis.com
- 14. mailto:eric.althoff@novartis.com
- 15. mailto:jeannie.neufeld@novartis.com
- 16. mailto:investor.relations@novartis.com